Author's response to reviews

Title: Analysis of cell cycle-related proteins in gastric intramucosal differentiated-type cancers based on mucin phenotypes: a new concept of molecular carcinogenesis for early differentiated-type gastric cancers

Authors:

Tamotsu Sugai (tsugai@cocoa.ocn.ne.jp)
Mitsunori Tsukahara (tsugai@iwate-med.ac.jp)
Masaki Endoh (msendo@iwate-med.ac.jp)
Yoshihiro Shioi (y-shioi@mta.biglobe.ne.jp)
Noriko Takebe (tnoriko@iwate-med.ac.jp)
Yoshiharu Mue (yo-mue0427@kih-biglobe.ne.jp)
Hiroo Matsushita (mattsu@mui.biglobe.ne.jp)
Minoru Toyota (mtoyota@sapmed.ac.jp)
Kazuyuki Suzuki (kasuzuki@iwate-med.ac.jp)

Version: 10 Date: 28 May 2010

Author's response to reviews: see over
Dear Dr. Norton:

Thank you very much for your e-mail of May 21, 2010, as well as the comments from the reviewers concerning our manuscript (MS no 1101811677312795). Herewith, I am sending you our two revised manuscripts, consisting of 26 pages of text (including 1 page of figure legends), 3 tables and 4 figures.

We have incorporated the reviewers' suggestions as follows:

**Reply to major revisions.**

1. We have revised Tables that we submitted, according to your suggestion.

2. We have ensured that title is identical in the manuscript and the submission system.

We believe that the revised manuscript addresses all of the major concerns of the Editorial Production Team and is now suitable for publication in the BMC Gastroenterology. We look forward to your favorable reply.

Yours sincerely,

Tamotsu Sugai, M.D.